Author's response to reviews

Title: Why do adult women in Vietnam take iron tablets?

Authors:

Ritsuko Aikawa (ritosuko_aikawa@hotmail.com)
Masamine Jimba (mjimba@m.u-tokyo.ac.jp)
Khan C Nguyen (nckhan@hn.vnn.vn)
Yun Zhao (Y.Zhao@exchange.curtin.edu.au)
Colin W Binns (C.Binns@curtin.edu.au)
Mi Kyung Lee (M.K.Lee@murdoch.edu.au)

Version: 6 Date: 1 June 2006

Author's response to reviews:

Dear Editor
WE have now completed the formatting changes requested. (Table legends deleted and table 5 is now cited)
Thankyou
MiKyung Lee